Author's response to reviews

Title: Factors associated with dental caries among institutionalized residents with schizophrenia in Taiwan: a cross-sectional study

Authors:

Kuan-Yu Chu (kyc0321@yahoo.com)
Nan-Ping Yang (g880703@ym.edu.tw)
Pesus Chou (pschou@ym.edu.tw)
Hsien-Jane Chiu (chiu8@mail2000.com.tw)
Lin-Yang Chi (chily@ym.edu.tw)

Version: 10 Date: 12 August 2010

Author's response to reviews:

Dear Prof. Titmus,

I would like to appreciate that BMC Public Health accepted the manuscript entitled" Factors associated with dental caries among institutionalized residents with schizophrenia in Taiwan- a cross-sectional study" (MS: 8899253593491398) for publication. We have checked this manuscript and complete the article payment. Thanks for your kind assistance.

Sincerely Yours,

Lin-Yang Chi, DDS, PhD
Associated professor
Department of Dentistry
National Yang-Ming University
Taiwan